

April 21, 2014

Dear Editor,

Pleased find enclosed the edited manuscript in Word format (file name: 7935. edited.doc)

**Title:** Contrast-enhanced harmonic EUS imaging: basic principles, present situation and future perspectives.

**Authors:** María-Victoria Alvarez-Sánchez, Bertrand Napoléon

**Name of journal:** World Journal of Gastroenterology

**ESPS Manuscript N°:** 7935

The manuscript has been improved according to the suggestions of reviewers:

- 1.- Format has been updated.
- 2.- Revision has been made according to the suggestions of the reviewer 734317:

Thank you for your comments. The manuscript has been edited and a language polishing performed.

3. - Primitive figure files are provided.

Thank you again for publishing our manuscript in *World Journal of Gastroenterology*.

Sincerely yours,

María Victoria Alvarez-Sánchez MD  
Gastroenterology Department - Endoscopy Unit  
Complejo Hospitalario de Pontevedra  
Avda. Mourente s/n. 360071 Pontevedra - Spain  
victoria.alvarez.sanchez @ hotmail.com